
diff -r cb2918e02806 -r b0ff6121cb9a handouts/ho02.tex
--- a/handouts/ho02.tex	Tue Sep 15 23:13:45 2020 +0100
+++ b/handouts/ho02.tex	Fri Sep 18 00:33:03 2020 +0100
@@ -13,15 +13,15 @@
 
 \section*{Handout 2 (Regular Expression Matching)}
 
-\noindent
-{\bf Checklist}
-
-\begin{itemize}
-  \item You have understood the derivative-based matching algorithm.
-  \item You know how the derivative is related to the meaning of regular
-  expressions.
-  \item You can extend the algorithm to non-basic regular expressions.
-\end{itemize}\bigskip\bigskip\bigskip
+%\noindent
+%{\bf Checklist}
+%
+%\begin{itemize}
+%  \item You have understood the derivative-based matching algorithm.
+%  \item You know how the derivative is related to the meaning of regular
+%  expressions.
+%  \item You can extend the algorithm to non-basic regular expressions.
+%\end{itemize}\bigskip\bigskip\bigskip
 
 \noindent
 This lecture is about implementing a more efficient regular expression




